Author's response to reviews

Title: Evaluation of QUADAS, a tool for the quality assessment of diagnostic accuracy studies

Authors:

Penny F Whiting (penny.whiting@bristol.ac.uk)
Marie E Westwood (mew3@york.ac.uk)
Anne W S Rutjes (a.rutjes@amc.uva.nl)
Johannes B Reitsma (j.reitsma@amc.uva.nl)
Patrick M M Bossuyt (p.m.bossuyt@amc.uva.nl)
Jos Kleijnen (jos@kleijnen.freeserve.co.uk)

Version: 5 Date: 10 February 2006

Author’s response to reviews:

This is a response to editorial comments not reviewer comments